Author's response to reviews

Title: Social Cohesion and Belonging Predict the Well-Being of Community-Dwelling Older People

Authors:

Jane M Cramm (cramm@bmg.eur.nl)
Anna P Nieboer (nieboer@bmg.eur.nl)

Version: 5
Date: 26 February 2015

Author's response to reviews:

1. Kindly rename your Introduction to Background.
   We renamed accordingly.

2. Please include a separate Conclusion Section.
   As requested we included a separate concluding section.